Author’s response to reviews

Title: Emotional and Behavioral Problems and associated factors among children and adolescents on Highly Active Anti-Retroviral Therapy in public hospitals of West Gojjam zone, Amhara Regional state of Ethiopia, 2018 : a cross sectional study

Authors:
Demewoz Kefale (demewozk@yahoo.com)
Abdisa Boka (bokaabdisa@yahoo.com)
Zureyash Mengstu (zuriyashaau@yahoo.com)
Zelalem Belayneh (zelalembe45@gmail.com)
Shegaw Zeleke (shegawzeleke@gmail.com)

Version: 2 Date: 05 Mar 2019

Author’s response to reviews:

Point by point response

Dear editor, thank you for your fast response and constructive comments. We have corrected the suggestions and submitted the requires supplementary files in the submission system. If additional inquiry is there please contact us freely.

Thank you

Regards

Demewoz Kefale

1. In our system, the title of your manuscript currently reads as “Authors’ response [REMOVED TABLE]”. Please revise your title in the submission system.

Reply: Thank you very much. The title is” Emotional and Behavioral Problems and associated factors among children and adolescents on Highly Active Anti-Retroviral Therapy in public hospitals of West Gojjam zone, Amhara Regional state of Ethiopia, 2018 : a cross sectional study”. But we could not find the difference in the system. If so please revert it accordingly.
2. Submission of a manuscript to a BioMed Central journal implies that materials described in the manuscript, including all relevant raw data, will be freely available to any scientist wishing to use them for non-commercial purposes, without breaching participant confidentiality. Thus, we request that you upload English language versions of all questionnaires developed specifically for use in this study as supplementary files. If the questionnaires are not yours or are already published by you, please supply references and/or links to them.

Reply: Thank you. We have uploaded the English language versions of all questionnaires developed specifically for use in this study as supplementary files.

3. Smaller tables may be inserted into the text or pasted at the end of the manuscript file after references. Tables larger than one A4/Letter page can be placed after the references or uploaded separately as supplementary files.

Reply: Thank you very much. We have submitted uploaded a table (table3) separately as supplementary files. Other tables are inserted into the text as they are smaller than one A4/Letter page.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Reply: We have submitted only the clean version of the revised manuscript.

Dear reviewer, thank you for the English language edition of our manuscript. We accept your corrections totally.

Thank you